Author’s response to reviews

Title: A new mouse model to study the role of ectopic Nanos3 expression in cancer

Authors:

Vanessa Andries (vanessa.andries12@gmail.com)
Evi De Keuckelaere (evidencekeuckelaere@gmail.com)
Katrien Staes (Katrien.Staes@irc.UGent.be)
Tino Hochepied (Tino.Hochepied@irc.ugent.be)
Joachim Taminau (Joachim.Taminau@irc.ugent.be)
Kelly Lemeire (Kelly.Lemeire@irc.ugent.be)
Philippe Birembaut (pbirembaut@chu-reims.fr)
Geert Berx (Geert.Berx@irc.UGent.be)
Frans van Roy (frans.vanroy@ugent.be)

Version: 2 Date: 28 May 2019

Author’s response to reviews:

Revised manuscript: BCAN-D-18-02739-R2

Our point-by-point response letter follows here (in capitals).

Editor Comments:

1. Please ensure that the Abstract in your manuscript file matches then submitted onto the submission system.

WE HAVE CHECKED THIS. THE ABSTRACT IN THE MANUSCRIPT MATCHES NOW THE ONE SUBMITTED ONTO THE SUBMISSION SYSTEM.

2. In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

THE REVISED MANUSCRIPT IS UPLOADED NOW AS A FINAL CLEAN VERSION. ALL RELEVANT TABLES/FIGURES/ADDITIONAL FILES HAVE ALSO BEEN UPLOADED AS CLEAN VERSIONS. ALL FIGURES, TABLES AND ADDITIONAL/SUPPLEMENTARY FILES ARE CITED WITHIN THE TEXT.

Reviewer reports:

WE ARE HAPPY THAT EACH OF THE THREE REVIEWERS RECOMMENDS NOW ACCEPTANCE WITHOUT FURTHER REVISION.